Author's response to reviews

Title: A Detailed Hapmap of the STSL locus spanning 40kb; Differences between Caucasians and African-Americans

Authors:

Bhaswati Pandit (bhaswati.pandit@mssm.edu)
Gwang-Sook Ahn (ahngs@dju.ac.kr)
Starr E Hazard (Hazard@musc.edu)
Derek Gordon (gordon@linkage.rockefeller.edu)
Shailendra B Patel (sbpatel@mcw.edu)

Version: 2 Date: 27 October 2005

Author's response to reviews:

Revised according your email to make it suitable for BMC Medical Genetics